Author’s response to reviews

Title: Assessing refugee healthcare needs in Europe and implementing educational interventions in primary care: a focus on methods

Authors:

Christos Lionis (lionis@galinos.med.uoc.gr)
Elena Petelos (elena.petelos@med.uoc.gr)
Enkeleint - Aggelos Mechili (mechili@uoc.gr)
Dimitra Sifaki-Pistolla (spdimil1@gmail.com)
Vasiliki-Eirini Chatzea (valchatz@hotmail.com)
Agapi Angelaki (a.angelaki@uoc.gr)
Imre Rurik (rurik.imre@sph.unideb.hu)
Danica Rotar-Pavlic (danica.rotar@gmail.com)
Christopher Dorwick (cfd@liverpool.ac.uk)
Michel Dückers (M.Duckers@nive1.nl)
Dean Ajdukovic (dajdukov@ffzg.hr)
Helena Bakic (hbakic@ffzg.hr)
Elena Jirovsky (elena.jirovsky@univie.ac.at)
Elisabeth-Sophie Mayrhuber (elisabeth.mayrhuber@meduniwien.ac.at)
Maria van den Muijsenbergh (Maria.vandenMuijsenbergh@radboudumc.nl)
Kathryn Hoffmann (kathryn.hoffmann@meduniwien.ac.at)

Version: 4 Date: 22 Jan 2018

Author’s response to reviews:

Point by point
Thank you for your kind words and valuable suggestions that we followed to improve our manuscript. Additional revisions are marked in the revised manuscript by track changes.

Editor Comments:

(1) Please accept my apologies that this was not requested from you earlier but, as your manuscript does not constitute what we would consider as a 'Debate' article in the journal, we will need to proceed with this as a 'Correspondence' article. Please view the guidelines here: http://www.biomedcentral.com/p/the-bmc-series-journals/correspondence and format your article according to these.

Response: We acknowledge all your efforts to enhance the manuscript and towards this direction, we have formatted the article as a “Correspondence article”.

(2) Please state in your 'Ethics approval and consent to participate' section the specific names of the ethics committees which approved your study.

Response: Thanks for your kind comment. We have added the requested information. Please find it at page 22, “Ethics approval and consent to participate section”, lines 510-517.

(3) Please format your 'Authors' Contributions' section as per the following and ensure to refer specifically to each author:

Response: We thank the editor for the kind comment. We have added the requested information. Please find it at page 23, “Authors' contributions” section, lines 548-551.